Author’s response to reviews

Title: Advancing Laboratory Medicine in Hospitals through Health Information Exchange: A Survey of Specialist Physicians in Canada

Authors:
Louis Raymond (louis.raymond@uqtr.ca)
Eric Maillet (eric.maillet@usherbrooke.ca)
Marie-Claude Trudel (marie-claude.trudel@hec.ca)
Josianne Marsan (josianne.marsan@sio.ulaval.ca)
Ana Ortiz de Guinea (ana.ortiz-de-guinea@hec.ca)
Guy Paré (guy.pare@hec.ca)

Version: 2 Date: 10 Feb 2020

Author’s response to reviews:

Editor's comment:
Based on these reports, and my own assessment as Editor, I am pleased to inform you that it is acceptable for publication in BMC Medical Informatics and Decision Making, once you have carried out one minimal adaptation that was suggested by one of our reviewers.

Authors' response:

Thanks for your positive assessment of our revised manuscript. Very much appreciated.

Reviewer 1's comments:

Authors have sufficiently addressed my comments from previous round of review. I only have one pedantic recommendation: change "Chi-square" to "Chi-squared"; that is the correct mathematical dictation of this statistic's name.

Authors' response:

Thanks for your positive remarks. “Chi-square” has been replaced by “Chi-squared”, as suggested.
Reviewer 2's comments:

Authors answered the comments from my initial revision. I have nothing to add and accept the publication.

Authors' response:

Thanks for your positive assessment of our revised manuscript. Very much appreciated.